Author’s response to reviews

Title: Prostatic Blue Nevus, an unusual histopathological diagnosis.

Authors:

Nelson Montalvo (nmontalvof@gmail.com)
Ligia Redroban (ligiredroban@gmail.com)

Version: 4 Date: 19 June 2013

Author’s response to reviews:

To: Journal Editorial Office
BioMed Central

The authors of the article “Prostatic Blue Nevus, an unusual histopathological diagnosis, correct the 6th reference:

BEFORE CORRECTION:

AFTER CORRECTION:

Looking forward for your favorable approval.

Best regards.
Dr. Nelson Montalvo. nmontalvof@gmail.com
Dra. Ligia Redroban.
Servicio de Patología.
Hospital Metropolitano.
Quito –Ecuador
patologia2@hmetro.med.ec